\begin{table}%t6
\caption{Afterglow properties of  54 XRRs/XRFS:  X-ray ToO observations (AX), time in days of the start of  ToO and  satellite which performed the observation,  optical afterglow detection (OT), time   of the OT detection and observed  magnitude,  Radio afterglow detection (RT) and  host galaxy detection.}\label{tab4}
\small%\centerline
{
{
\begin{tabular}{lcccccc}
\hline\hline
Events & A X & ToO Date[$d$] and istrument &OT& Date[$d$] and $m_R$[$m$] OT & Radio T & Host galaxy  \\
\hline
\object{XRF 971019}   &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRR 971024}   &   --      & --                  &    --       &     --         &     --    &    --     \\
\object{XRR 980128}   &   --      & --                  &     --      &     --         &     --    &    --     \\
\object{XRR 980306}   &   --      & --                  &      --     &      --        &     --    &    --     \\
\object{XRF 981226}   & Y$^{1}$        &   0.3, 7.2 ($\sax$)  &  N$^{2}$       & 0.40  $m >23$ & Y$^{3}$        & Y?$^{4}$  \\
\object{XRF 990520}   & --        & --                   &  N$^{5}$         & 0.74 $m>22.5 $ &  N  $^{6}$     &    --      \\
\object{XRF 990526}   &  --       & --                  &     --      &      --        &     --    &    --      \\
\object{XRF 990704}   & Y $^{7}$       & 0.3, 7.1 ($\sax$)    &  N $^{8}$        & 0.19 $ m>22.5$  &  N $^{9}$      &  --       \\
\object{XRR 991106}   & Y? $^{10}$       & 0.33  ($\sax$)    &  N $^{11}$        & 0.42  $m>22$  &  N $^{12}$      &  --        \\
\object{XRF 000206}   &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRR 000208}   &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRF 000416}   &  --       & --                  &     N$^{13}$      & 2.64 $m>20.7$        &     --    &    --     \\
\object{XRF 000615}   & Y$^{14}$        & 0.42   ($\sax$)    &  N$^{15}$         & 0.18  $m>21.5$  &  --       &  --      \\
\object{XRF 010213}   & --        & --                  &  N $^{16}$        & 1.98  $m>22$  &  N  $^{17}$     &    --   \\
\object{XRF 010225}    &   --      & --                  &      --     &      --        &     --    &    --     \\
\object{XRR 010326B}   &   --      & --                  &      N$^{18}$     & 0.5 $m>21$             &     --    &    --     \\
\object{XRR 010613}     &   --      & --                  &      --     &      --        &     --    &    --     \\
\object{XRR 010629}     &   --      & --                  &      N $^{19}$    & 0.49  $m>20.5$             &     --    &    --     \\
\object{XRR 010921}      &   --      & --                  &      Y$^{20}$     & 1.05 $m=19.9\pm0.2$             &     --    &    Y    \\
             &   --      & --                  &          & 0.92  $m>20.5$  $^{21}$           &     --    &    --     \\
\object{XRF 011019}       &   --      & --                  &      N$^{22}$     & 1.07  $m>22.5$             &     --    &    --     \\
\object{XRR 011030}       & Y $^{23}$       & 11, 31.2    ($\chandra$)    &  N$^{24}$         & 0.30  $m>21$  &  Y   $^{25}$    &  Y?$^{26}$  \\
\object{XRR 011103}       &   --      & --                  &      --     &      --        &     --    &    --     \\
\object{XRF 011130}       & Y $^{27}$       & 10, 83    ($\chandra$)    &  Y? $^{28}$        & 6.89  $m=23.0\pm0.1$  &  Y? $^{29}$       &  Y  $^{30}$\\
\object{XRR 011211}       & Y $^{31}$       & 0.5    ($\xmm$)    &  Y $^{32}$        & 0.46  $m=20.41\pm0.04$   &  --     &  Y  $^{33}$\\
\object{XRF 011212}       &   --      & --                  &      Y?$^{34}$     & 2.00  $m=23.8\pm0.2$             &     --    &    --     \\
\object{XRR 020124}       &   --      & --                  &      Y $^{35}$    & 1.51 $m=23.84\pm0.17$              &     --    &    --     \\
\object{XRR 020127}       & Y   $^{36}$& 4.1, 14.6     ($\chandra$)    &  N  $^{37}$        & 0.19  $m>19.5$  &  Y  $^{38}$      &  Y$^{39}$   \\
\object{XRF 020317}       &   --      & --            &      Y?  $^{40}$    & 0.76  $m=19.6\pm0.1$             &     --    &    --     \\
\object{XRR 020410}       & Y  $^{41}$       & 0.83, 2.3     ($\sax$)    &  Y  $^{41}$         & 0.26  $m=21.0\pm0.5$  &  N  $^{42}$&  --           \\
\object{XRF 020427}       & Y $^{43}$        & 0.46, 2.5 ($\sax$)    &  --        & --  &  Y? $^{44}$       &  Y? $^{45}$            \\
             &          & 9.2, 17.2($\chandra$)  &         &         &    &     \\
\object{XRF  020625}    &   --      & --                  &      N $^{46}$     & 0.44   $m>18.2$             &     --    &    --     \\
\object{XRR 020812}    &   --      & --                  &      N  $^{47}$    & 0.13  $m>19$             &     --    &    --     \\
\object{XRR 020819}    &   --      & --                  &      N  $^{48}$    & 0.72  $m>21.7$             &     Y  $^{49}$   &    Y? $^{50}$ \\
\object{XRF 020903}    &   --      & --                  &      Y $^{51}$     & 26.5   $m=18.6\pm0.2$&    Y? $^{52}$   &    Y  $^{53}$   \\
          &         &                   &            & 1.00  $m>19.8$ $^{54}$  &     & \\
\object{XRR 021004}      & Y$^{55}$        & 0.85, 52.3($\chandra$)    &  Y$^{56}$& 0.48  $m=18.46\pm0.05$   &  Y$^{57}$&  Y$^{58}$     \\
\object{XRF 021021}    &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRF 021104}     &   --      & --             & N $^{59}$    & 0.12  $m>21$             &     --    &    --     \\
\object{XRR 021112}   &   --      & --                  &      N $^{60}$    & 0.08  $m>21$             &     N$^{61}$    &    --     \\
\object{XRR 021211}    &   --      & --                  &      Y $^{62}$    & 0.85  $m=23.20\pm0.18$             & N $^{63}$   & Y$^{64}$      \\
\object{XRR 030115}    &   --      & --                  &      N$^{65}$     & 0.48  $m>21$             &Y $^{66}$  &Y$^{67}$     \\
\object{XRR 030323}   &   --      & --                  &   Y $^{68}$   & 0.35  $m=18.83\pm0.09$             & --   &  Y$^{69}$      \\
\object{XRR 030324}    &   --      & --                  &      N$^{70}$     & 0.39  $m>21.4$             &     N$^{71}$    &Y?$^{72}$     \\
\object{XRR 030329}     & Y$^{73}$ & 0.2, 1.3    ($\rxte$)    &  Y $^{74}$        & 0.37  $m=14.55\pm0.03$  &  Y $^{75}$ &  Y$^{76}$       \\
             &          & 37, 61, 258($\xmm$)  &         &         &    &     \\
\object{XRF 030416}  &   --      & --                  &      N$^{77}$     & 1.45  $m>15$             &     --    &    --     \\
\object{XRR 030418} &   --      & --                  &      Y$^{78}$     & 6.58  $m=24.9\pm0.4$             &    --     &   --      \\
\object{XRF 030429}  &   --      & --                  &      Y$^{79}$     & 0.37  $m=20.20\pm0.15$             & N$^{80}$         &    Y$^{81}$  \\
\object{XRF 030528}  & Y$^{82}$        & 6, 12    ($\chandra$)    &  N$^{83}$       & 0.002  $m>18.7$  &  --       &  Y $^{84}$       \\
\object{XRF 030723}  & Y $^{85}$       & 2.1, 12.7     ($\chandra$)    &  Y $^{86}$        & 0.86  $m=21.9\pm0.2$  &  N$^{87}$      &  --      \\
\object{XRR 030725}    &   --      & --                  &      Y$^{88}$     & 3.89  $m=21.2\pm0.2$             &   --    &  --    \\
\object{XRR 030821}    &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRF 030823}   &   --      & --                  &      N$^{89}$     & 0.78  $m>22.5$             &     --    &    --     \\
\object{XRF 030824}    &   --      & --                  &      N$^{90}$     & 1.72  $m>22.5$             &     --    &    --     \\
\object{XRR 031109B}   &  --       & --                  &     --      &      --        &     --    &    --     \\
\object{XRR 031220} & Y$^{91}$        & 5.6, 28.5   ($\chandra$)    &  N$^{92}$         & 0.23  $m>21$  &  --       &  --          \\
\hline
\end{tabular}}}
\medskip
 (1): \citet{fro00},   (2): \citet{lind},   (3): \citet{frail}, (4): \citet{holland}, (5): \citet{ct},  (6): \citet{frail2},
(7): \citet{fero01}, (8): \citet{jens}, (9): \citet{rol}, (10): \citet{anto},  (11): \citet{jensb}, (12): \citet{frailB},   (13): \citet{price},  (14): \citet{nic01},  (15): \citet{stan00},   (16): \citet{zhu},  (17): \citet{ber01A},  (18): \citet{price4},  (19): \citet{and},   (20): \citet{priceb, lamb01, bloom1},  (21): \citet{ser}, (22): \citet{kom01},    (23): \citet{heise},  (24): \citet{moh}, (25): \citet{harr}, (26): \citet{blooma},  (27): \citet{ricker, but},  (28): \citet{gar}, (29): \citet{ber01B},   (30): \citet{gar}, (31): \citet{piro04}, (32): \citet{ses},    (33): \citet{bur},   (34): \citet{dul}, (35): \citet{gor}, (36): \citet{fox},  (37): \citet{lamb02},  (38): \citet{rol2}, (39): \citet{fox2},
(40): \citet{tom}, (41): \citet{nic04}, (42): \citet{frail3},  (43): \citet{ama04}, (44): \citet{wi},  (45): \citet{ct2,blooma}, (46): \citet{burn},
(47): \citet{kaw},   (48): \citet{pic02}, (49): \citet{hen},   (50): \citet{lev},  (51): \citet{ste},  (52): \citet{ber02A},  (53): \citet{lev2}, (54): \citet{price2},  (55): \citet{rsak2, rsak1}, (56): \citet{sat},  (57): \citet{frail02},  (58):  \citet{dostoj2}, (59): \citet{foxP2},
(60): \citet{sch},              (61): \citet{frailBer},  (62): \citet{mac02}, (63): \citet{ber2F},   (64): \citet{fruch2}, (65): \citet{ny},
(66): \citet{ber2F3},    (67): \citet{mas03},   (68): \citet{woo}, (69):  \citet{vreeswijk},  (70):  \citet{ric2},  (71): \citet{ber2F34},  (72): \citet{ny4},
(73): \citet{marshall,  marshall2, tiego2, schartel, tiego3}, (74): \citet{bar},   (75): \citet{ber223},     (76): \citet{bloomB},  (77): \citet{lip},    (78): \citet{dul3},
(79): \citet{rum}, (80): \citet{ber2F343},  (81): \citet{kam},  (82): \citet{but3}, (83): \citet{ay}, (84): \citet{rau04}, (85): \citet{but3c, but3b},   (86): \citet{fynbo},
(87): \citet{fynbo3},  (88): \citet{vin}, (89): \citet{foxP}, (90): \citet{foxP1}, (91): \citet{gen04},  (92): \citet{fox3}.
\end{table}